Author's response to reviews

Title: Road traffic accident mortality in Vietnam: Evidence for policy from a national sample mortality surveillance system

Authors:

Anh D Ngo (anh.ngo@unisa.edu.au)
Chalapati Rao (c.rao@sph.uq.edu.au)
Nguyen Thi Phuong Hoa (hoatuber@yahoo.com)
Damian Hoy (d.hoy@uq.edu.au)
Khieu Thi Quynh Trang (sep9th80@yahoo.com)
Peter S Hill (p.hill@uq.edu.au)

Version: 3 Date: 4 January 2012

Author's response to reviews:

Dear Jimmar Dizon,

Thank you for your email. We have made changes according to your comments as below:

1. Experimental research that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/e/policy/b3.htm). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

We have added a sentence on page 5 paragraph 1 in the method section that: “Ethical approval was given by the Ethic Committee Hanoi Medical University, Vietnam”.

2. Please include the name of the group within the list of authors, by adding the name to the 'first name' box; and the bracketed abbreviation to the 'last name' box in the submission system. Please remove the names of the members of the group from the title page and place them in the Acknowledgements, as they are to be acknowledged as members of the study group. The groups' contribution as a whole can be given in the authors contributions section. Please explicitly mention each author.

We have removed the names of authors from the title page to the Acknowledgments.

Kind Regards,

Anh D Ngo
Corresponding author